Author's response to reviews

Title: Phospho-specific flow cytometry identifies aberrant signaling in indolent B-cell lymphoma

Authors:

Egil S Blix (egil.blix@uit.no)
Jonathan M Irish (jonathan.irish@gmail.com)
Anne Husebekk (Anne.Husebekk@unn.no)
Jan Delabie (jan.delabie@medisin.uio.no)
Lise Forfang (Lise.Forfang@rr-research.no)
Anne Tierens (ATN@ous-hf.no)
June H Myklebust (June.Helen.Myklebust@rr-research.no)
Arne Kolstad (ARNEK@ous-hf.no)

Version: 2 Date: 10 October 2012

Author's response to reviews: see over
Dear Editor

Thank you for the comments from Associate Editor and referee.

Please find the revised manuscript entitled “Phospho-specific flow cytometry identifies aberrant signaling in indolent B-cell lymphoma” enclosed for your consideration.

This cover letter gives a point-by-point response to the concerns from referee and Associate Editor. All changes made in the manuscript are highlighted with tracked changes.

- On page 1 there is an alteration in two of the author’s affiliation.
- On page 5 we have corrected the sentence describing the function of ITAMs.
- On page 9 and 10 we have omitted the BCL-2 antibody from methods since BCL-2 data are not presented in this study.
- On page 10 in methods we provide a more detailed description of fluorescent cell barcoding.

Please contact me if you have questions regarding the manuscript.

Yours sincerely,

Egil Blix, MD